

Hyp<u>E</u>rpolarised <u>Xenon Magnetic Resonance PuLmonary ImAging In</u> PatieNts with Long-COVID (EXPLAIN)

Ethics Ref: 21/SC/0398 IRAS Project ID: 305846 Date and Version No: 13Apr2022, V3.0

- Chief Investigator: Professor Fergus Gleeson¹
- Investigators:

Professor Najib Rahman¹

Dr Andrew Swift²

Professor Jim Wild²

Professor Ly-Mee Yu¹

- Dr Betty Rahman¹
- Dr James Grist¹

Professor Allan Lawrie²

Dr Roger Thompson²

Dr Emily Fraser³

Dr Ling-Pei Ho¹

Dr Helen Davies⁴

Professor Alex Horsley⁵

Affiliations:

- 1. University of Oxford
- 2. University of Sheffield
- 3. Oxford University Hospitals NHS Foundation Trust
- 4. University Hospital of Wales
- 5. University of Manchester

Sponsor:	University of Oxford Joint Research Office 1st floor, Boundary Brook House Churchill Drive Headington Oxford OX3 7GB E-mail: ctrg@admin.ox.ac.uk
Funders:	Cohort A: National Consortium of Intelligent Medical Imaging, University of Oxford Cohorts B & C: National Institute of Health Research (NIHR)

Chief Investigator Signature:

P. Glum Victoria thrio

Statistician Signature:

Disclosures

None

Confidentiality Statement

This document contains confidential information that must not be disclosed to anyone other than the Sponsor, the Investigator Team, HRA, host organisation, and members of the Research Ethics Committee, unless authorised to do so.

TABLE OF CONTENTS

1.	KEY CONTACTS
2.	LAY SUMMARY7
3.	SYNOPSIS
4.	ABBREVIATIONS
5.	BACKGROUND AND RATIONALE
6.	OBJECTIVES AND OUTCOME MEASURES 16
7.	STUDY DESIGN 16
8.	PARTICIPANT IDENTIFICATION
8.1.	Study Participants 18
8.2.	Inclusion & Exclusion Criteria 20
9.	PROTOCOL PROCEDURES
9.1.	Recruitment 22
9.2.	Screening and Eligibility Assessment
9.3.	Informed Consent
9.4.	Randomisation
9.5.	Blinding and code-breaking
9.6.	Description of study intervention(s), comparators and study procedures (clinical) 24
9.	6.1 Description of study intervention(s)
9.	6.2 Description of comparator(s)
9.	6.3 Description of study procedure(s)24
Dem	nographics, past medical and smoking (10 mins)24
Obs	ervations (10-15mins)
Lung	g Function Test (30 Mins)
Que	stionnaires (45mins)
Low	dose CT scan
ΗРХ	C-pMRI or HPX-MRI (approximately 30mins)25
Gad	olinium contrast
Card	diac MRI
9.7.	Baseline and follow up assessments

Initial	patient approach	
Visit ⁻	1 Baseline (Participants post COVID-19 infection)	
Visit 2	2 and 3 (3-12 months post baseline visit)	
9.8.	Early Discontinuation/Withdrawal of Participants	27
9.9.	Definition of End of Study	27
10. SA	FETY REPORTING	27
10.1.	Definition of Serious Adverse Events	27
10.2.	Reporting Procedures for Serious Adverse Events	28
11.ST	ATISTICS AND ANALYSIS	
11.1.	Description of the Statistical Methods	
11.2.	Sample Size Determination	29
11.3.	Analysis populations	29
11.4.	Dissemination of Findings	29
11.5.	Stopping rules	30
11.6.	The Level of Statistical Significance	30
11.7.	Procedure for Accounting for Missing, Unused, and Spurious Data.	30
11.8.	Procedures for Reporting any Deviation(s) from the Original Statistical Plan	30
11.9.	Health Economics Analysis	30
12. DA	ATA MANAGEMENT	30
12.1.	Source Data	31
12.2.	Access to Data	31
12.3.	Data Recording and Record Keeping	31
12.4 Da	ata Sharing	32
13. QL	JALITY ASSURANCE PROCEDURES	32
13.1.	Risk assessment	32
13.2.	Study Monitoring	33
13.3.	Study Committees	33
13.3.	1 Trial Management Group	33
13.3.2	2 Trial Steering Committee	33
13.3.3	3 ORTU Safety Oversight Group (SOG)	33
14. PR	ROTOCOL DEVIATIONS	33

15. SEI	RIOUS BREACHES	33
16.ETH	HICAL AND REGULATORY CONSIDERATIONS	33
16.1.	Declaration of Helsinki	
16.2.	Approvals	
16.3.	Other Ethical Considerations	
16.4.	Reporting	
16.5.	Transparency in Research	
16.6.	Participant Confidentiality	
16.7.	Expenses and Benefits	
17. FIN	IANCE AND INSURANCE	35
17.1.	Funding	35
17.2.	Insurance	35
17.3.	Contractual arrangements	35
18. PUI	BLICATION POLICY	35
19. DE ^v PROPI	VELOPMENT OF A NEW PRODUCT/ PROCESS OR THE GENERATION OF INT ERTY	ELLECTUAL
20. AR	CHIVING	35
21.REI	FERENCES	
22. API	PENDIX B: SCHEDULE OF STUDY PROCEDURES	38
23. API	PENDIX C: AMENDMENT HISTORY	42

1. KEY CONTACTS

Chief Investigator	Professor Fergus Gleeson				
	fergus.gleeson@oncology.ox.ac.uk				
Sponsor	University of Oxford				
	Joint Research Office, Boundary Brook House				
	Headington, Oxford, OX3 7GB				
	Office: +44 01865 616480				
Funder(s)	Cohort A: National Consortium of Intelligent Medical Imaging,				
	University of Oxford				
	Cohorts B & C: National Institute of Health Research (NIHR)				
Clinical Trials Unit	Oxford Respiratory Trials Unit				
Statistician	Professor Ly-Mee Yu				
Committees	Trial Management Group				
	Trial Steering Committee				
	ORTU Safety Oversight Group				
	The Chair of the TMG will be the Chief Investigator of the trial.				
	The Chair of the TSC will be Simon Padley, Royal Brompton Hospital				
	All of the above committee details are within the group/committee				
	charters.				

2. LAY SUMMARY

We wish to understand why some individuals with Long-COVID struggle with breathlessness on exertion (when active) and have a reduced ability to exercise. To do this, we will use MRI scanning and a special gas (hyperpolarised xenon) which is breathed in during the scan. The xenon gas is harmless in the quantity we use. This technique shows the movement of xenon within the lungs and moving out of the lungs into the bloodstream, similar to how oxygen is absorbed. In patients hospitalised with COVID-19, we found that the xenon MRI scans several months after discharge showed lung damage, even when other tests were normal. Importantly, on follow-up imaging, some have remained abnormal.

We will study more post-hospitalised patients with Long COVID (Cohort A) to determine if our initial findings are confirmed in a larger number of patients. We will also use the same scan technique in individuals with long-COVID not admitted to hospital (Cohort B), who have ongoing breathlessness on exertion to see if they have lung damage. If lung abnormalities are found, we will assess how severe they are, whether they provide an explanation for the ongoing breathing problems, and whether they improve over time. To do this, we will compare scans in individuals with long-COVID who are breathless on exertion (Cohort B, Groups 3 & 4) to those without breathlessness as a symptom (Cohort B, Group 5), and to patients infected by the virus but didn't have any symptoms (Cohort C), enabling us to see if breathing problems are related to lung damage.

We will also give intravenous contrast to look to see if the lung problems are due to blood clots, and also perform a cardiac MRI scan at the same time in some of the participants attending for their Xenon MRI scans. This sequence has already been studied in another research study, and we will learn from this and use it along with our Xenon scans to identify if there is cardiac damage either as the cause of breathlessness or associated with abnormalities seen on the Xenon scans.

Our aim is to further our understanding of some of the factors that cause symptoms in Long-COVID, and provide a much needed explanation to individuals struggling with breathlessness. Learning more about the nature of damage within the lungs through xenon MRI may help with the future development of treatments, and provide a reliable way of measuring the treatment response over time.

3. SYNOPSIS

In this study we aim to identify and characterise the extent of lung and possibly cardiac damage in patients attending Long COVID clinics, whether they were infected in the first wave in the Spring of 2020, or in subsequent waves.

Study Title	HypErpolarised Xenon Magnetic Resonance PuLmonary Imaging in PAtieNts with Long COVID
Short title	EXPLAIN
Study registration	ISRCTN TBC
Sponsor	University of Oxford
	Joint Research Office
	1 st floor, Boundary Brook House
	Churchill Drive
	Headington
	Oxford OX3 7GB
	E-mail: <u>ctrg@admin.ox.ac.uk</u>
Funder	Cohort A: National Consortium of Intelligent Medical Imaging, University of Oxford
	Cohorts B & C: National Institute of Health Research
Study Design	Prospective observational cohort study
Study Participants	Cohort A: Post-hospitalisation cohort (100)
	100 Hospitalised [*] Long COVID patients, with normal or near normal CT scans
	Cohort B: Non-hospitalised cohort (250)
	200 participants - Non-Hospitalised Long COVID (NHLC) clinic attendees with
	symptoms of breathlessness
	50 participants - Non-Hospitalised Long COVID (NHLC) clinic attendees with no
	symptoms of breathlessness
	Cohort C: Controls (50)
	50 control participants with a positive COVID test but no symptoms of Long-COVID
	Hospitalised refers to participants that required hospital management of COVID-19 infection.
Sample Size	400 participants

Planned Study	Planned start date: 01 Nov 2021			
Period	Planned end date: 31 Oct 2023			
	Total project duration: 24 months			
Planned	18 months			
Recruitment				
period				
	Objectives	Outcome Measures	Timepoint(s)	
Primary	To evaluate, using HPX-	a) The detection of	a) Baseline scan	
	pMRI whether	diffusion and/ or		
	unexplained exertional	perfusion defects		
	dyspnoea in NHLC is due			
	to lung damage			
		b) The determination of	b) Baseline and	
		whether detected	follow-up scan	
		abnormalities correlate		
		with symptoms of		
		breathlessness		
		c) The degree of	c) End of study	
		pulmonary damage and		
		change detected on		
		follow up scanning		
Casandan	To determine if there are	The detection of condice	Deseline seen	
Secondary		MBL observe of the	Baseline scan	
		MRI abnormaillies		
	detected pulmonary			
	abnormalities or present			
	alone as a cause for NHLC			
	patient breathlessness			
Intervention(s)	Not applicable			
Comparator	Not Applicable			

4. ABBREVIATIONS

ACE-2	Angiotensin converting enzymes-2
CI	Chief Investigator
CFS	Clinical Frailty Survey
CMR	Cardiovascular Magnetic Resonance
CoV	Coronavirus
COVID-19	Coronavirus disease 19
CRF	Case Report Form
СТ	Computed Tomography
ECG	Electrocardiograph
eGFR	Estimated Glomerular Filtration Rate
EQ5D-5L	EuroQol- 5 Dimension 5 Level descriptive system
FACIT-F	The Functional Assessment of Chronic Illness Therapy – Fatigue Survey
GAD-7	Generalised Anxiety Disorder survey
GCP	Good Clinical Practice
GP	General Practitioner
HPX	Hyperpolarised Xenon
HPX-pMRI	Hyperpolarised Xenon Perfusion Magnetic Resonance Imaging
HRA	Health Research Authority
ICF	Informed Consent Form
IEP	Image Exchange Portal
ILD	Interstitial lung disease
LC	Long COVID
LFT	Lung Function Test
mBORG	Modified BORG Score
MRI	Magnetic Resonance Imaging
MRS	Magnetic Resonance Spectroscopy
NCIMI	National Consortium of Intelligent Medical Imaging
NHLC	Non-Hospitalised Long COVID
NHS	National Health Service
ORRU	Oxford Radiology Research Unit
ORTU	Oxford Respiratory Trials Unit
PI	Principal Investigator
PIL	Participant/ Patient Information Leaflet
PHQ	Personal Health Questionnaire
RBC:TP	Red Blood Cell/ Tissue Plasma ratio
R&D	NHS Trust R&D Department
REC	Research Ethics Committee

RGEA	Research Governance, Ethics & Assurance, University of Oxford
SARS-CoV	Sudden Acute Respiratory Distress Syndrome Coronavirus
SOP	Standard Operating Procedure
SAE	Serious Adverse Effects
TTE	Transthoracic Echocardiography
Xe	Xenon
6MWT	6-minute walk test

5. BACKGROUND AND RATIONALE

Scientific background

COVID-19 pneumonia has specific features that differ from other respiratory viral infections. Acute COVID-19 pneumonia causes an unusual pattern of lung injury giving rise to early and profound hypoxaemia that is often refractory to oxygen therapy. This clinical picture is suggestive of excessive "shunting" of blood through poorly ventilated regions of the lung from which oxygen cannot be extracted, and of severe ventilation-perfusion mismatch. This may be caused by dysregulation of pulmonary vascular perfusion due to direct damage of the pulmonary microvasculature by the SARS-CoV-2 virus and/or the presence of multiple thrombi. Evidence for this includes radiological reports of abnormal pulmonary vessels and lung histology that demonstrates viral particles within pulmonary endothelium alongside destruction of the vasculature and the presence of microthrombi.

Hospitalised patients with severe COVID-19 pneumonia may develop persisting interstitial abnormalities within the areas of acute abnormality. Where this is the case, exercise limitation and exertional dyspnoea are not surprising. However, nearly all NHLC patients have normal CT scans with no structural cause for their symptoms identifiable. Dual energy CT (DECT) has been reported to identify perfusion defects due to microthrombotic disease in acute COVID-19 infection but is also nearly always normal in NHLC. The Oxford University Hospitals NHS Foundation Trust Radiology department has identified only one positive scan in a local case review of 100 patients with LC. Pulmonary function tests assume homogeneity within the lungs, giving a 'readout' for the entire lung, and hence are often "normal" in this population. We hypothesise that an investigation which provides regional data on the function of the alveolar/capillary interface (i.e. anatomical and functional for gas exchange) is required to delineate damage from COVID-19, or to thoroughly exclude the lung as the site of pathology in NHLC.

Proof of principle for our hypothesis is provided by the early published results from the Oxford HPX-MRI study - Post COVID-19 disease follow-up imaging using hyperpOlariSed xenon MRI and CT (C-MORE POST), which has shown that there are specific defects in gaseous diffusion across the alveolar epithelial-capillary membrane in post-hospitalisation Long COVID-19 patients, despite a normal CT scan⁴⁵. These findings imply ongoing impairment in ventilation-perfusion matching and diffusion impairment due to persistent damage of the alveolar-capillary membranes in the lungs during the recovery phase of the illness. This may partially explain the observation that survivors (including never-hospitalised patients) have noted a prolonged recovery characterized by breathlessness, persistent fluctuations in oxygen saturation and functional limitation. Hyperpolarised xenon MRI (HPX-MRI) is a state-of-the art imaging modality that is safe, reproducible and quantitative. It does not involve radiation exposure and is independent of patient respiratory effort. It enables sensitive regional investigation of the lung parenchyma and measures pulmonary ventilation and gas transfer across the alveolar epithelium into the capillaries and red blood cells.

The ability of Xenon to dissolve through the alveolar epithelium into the lung parenchyma and capillary blood may enable detection of changes in Long COVID-19 which is not possible using other imaging

techniques. The HPX-MRI report from Wuhan demonstrated abnormal gas-exchange detected by HPX-MRI Chest in post-hospitalised COVID-19 patients soon after discharge. In dyspnoeic patients following hospitalisation with COVID-19, diffusion abnormalities have been identified at 3-6 month after discharge, and persist in a proportion 6-12 months later (Oxford and Sheffield on-going prior trial data, which is unpublished) including non-ventilated patients. CT scans in these patients are normal or near normal indicating that this modality is not sensitive enough to pick up subtle abnormalities in the lung microstructure and/or pulmonary vasculature. Additionally, 9 Oxford patients have been scanned on two occasions, 3 months apart, and some have not shown significant improvement in the degree of diffusion impairment. We have also performed a few HPX-MRI Chest in NHLC patients as part of our pilot NHLC study, and some of them have shown dissolved phase defects similar to the post-hospitalisation patients.

Long-COVID is a complex and poorly understood syndrome with more than 200 symptoms described by sufferers. It is similar in some respects to Post-Viral Chronic Fatigue Syndrome (CFS), although there is one particularly interesting distinction; 80% of patients report breathlessness as a primary symptom. This is notably absent from prior descriptions of CFS. The mechanisms driving breathlessness and exertional dyspnoea in Long-COVID remain unclear and conventional investigations are typically unremarkable.

The mechanisms underlying the symptoms of Long-COVID are likely to be multiple, overlap and vary between patients. In patients who experience breathlessness at rest or suffer from acute recurrent episodes not related to exertion, a Breathing Pattern Disorder is commonly identified. Physiotherapy can help restore a normal breathing pattern and has been successful in alleviating symptoms in many. There are also a significant subset of Long-COVID sufferers who have reproducible breathlessness on exertion that limits daily activities. In these patients, we question whether exertional dyspnoea may be secondary to persistent abnormalities caused by loss of integrity or low-grade inflammation around the alveolar membrane and/or pulmonary vasculature. Evidence for this includes: 1) The HPX-MRI Chest findings in the post-hospitalised patients as described above, 2) Local data demonstrating that lung function tests, although usually normal, frequently show a gas transfer (a marker of alveolar and pulmonary vascular integrity) disproportionately lower than the accompanying lung volumes, 3) A proportion of patients exhibit evidence of oxygen desaturation on exertion, 4) The recently published C-MORE and CoverScan studies demonstrating MRI anomalies in multiple organs following COVID-19 infection, indicating that the virus can cause long-lasting changes, 5) Acute COVID-19 infection is highly prothrombotic, with up to 30% of those admitted to critical care found to pulmonary embolic disease on CT Pulmonary Angiograms. Furthermore, autopsy specimens have demonstrated the presence of widespread microthrombi within the pulmonary vasculature, and these vascular abnormalities may account for exercise intolerance.

The management of Long-COVID requires a holistic and multidisciplinary approach. However, the proportion of patients who struggle and are functionally incapacitated by breathlessness is significant, and it is frequently breathlessness that limits return to normal activity. As such, there is an urgent and

unmet need to understand the biological factors underpinning dyspnoea and exercise intolerance. Investigations such as HPX-MRI Chest combined with perfusion MRI Chest (HPX-pMRI – this involves injection of gadolinium contrast) provides highly sensitive and integrated information about the structure and function of the lungs and pulmonary vasculature that cannot be obtained through conventional investigations. Identification of abnormalities may potentially 1) impact on therapeutic strategies to potentially limit or improve this complication in those suffering with post-COVID syndrome 2) provide pathophysiology insights of disease and 3) provide information of the prevalence of pulmonary damage and the medium to long term burden of disease post COVID-19 infection.

Currently, patients diagnosed with Long-COVID attending specialist clinics commonly have no identifiable cause for their symptoms. Patients often undergo numerous investigations for exertional breathlessness, including CXRS, CT scans, lung function and echocardiography, which are normal in the vast majority. Over-investigation can heighten anxiety and exacerbate the psychological interpretation of symptoms; a single investigation that may provide detailed granular information about the function and structure of the lungs and pulmonary vasculature would supersede the need for numerous less sensitive tests.

We have shown in post-hospitalised patients that HPX-MRI Chest scans identify abnormalities when all other tests are non-informative. Determining whether diffusion impairments also exist in NHLC patients with exertional breathlessness will advance our understanding of some of the biological mechanisms driving Long-COVID symptomatology. It will help provide an explanation to sufferers frustrated at the lack of current answers and may potentially enable their better longer-term management. Conversely, HPX-pMRI Chest, if normal, will provide reassurance to patients that they have no lasting lung damage and this may also facilitate rehabilitation. HPX-pMRI Chest may potentially provide insight into the nature and trajectory of NHLC abnormalities and might also be used as a method to assess the effects of proposed medical interventions - preliminary discussions have been had with the HEAL-COVID investigators.

All patients recruited will have a low dose CT scan which will be scored as per our Radiology paper; this should be normal or nearly normal, as determined by study investigators, for them to be recruited (patients with post-COVID-19 Interstitial Lung Disease are being recruited into the NIHR funded UKILD-PCF study). Low dose CT has been shown to reduce radiation exposure, enabling imaging to be repeated if required to assess evolving changes, and adequately depicts interstitial lung disease. It is also now possible to differentiate and quantify CT changes, and to subdivide the changes identified. We will be able to co-register the CT and HPX-pMRI Chest scans, as both will be performed at approximately 1 litre inspiration.

Effects of COVID-19 on the heart

The relationship between COVID infection and cardiac disease is also important. Data from at least four major clinical studies of COVID-19 patients from China suggest that evidence of myocardial injury can be detected in 7-20%. Defined by an increase in cardiac biomarkers above the 99th percentile of normal

range or a new abnormality on electrocardiography or echocardiography, cardiac injury was noted to be far more common in patients (up to 46 % of cases) with more severe disease and non-survivors. Conversely, mortality was higher in patients with pre-existing cardiovascular disease or risk factors for coronary disease who contracted COVID-19. These findings suggest that cardiac complications are common and associated with disease severity. A number of mechanisms have been suggested to promote myocardial injury in SARS-CoV-2 infections. These range from a type of fulminant myocarditis, caused by cytokine storm, to infectious viral myocarditis and acute coronary syndrome. In support of a reactive myocarditis, several studies have demonstrated an acute rise in inflammatory cytokine levels and C-reactive protein, which is seen to precede the rise in serum biomarkers of myocardial injury. Experimental rabbit models have also demonstrated that SARS COV-2 can directly infect cardiomyocytes with the help of angiotensin converting enzymes-2 (ACE-2) receptors, raising the possibility of a viral myocarditis. Another potential cause for myocardial injury is an acute coronary syndrome due to plaque rupture. A post-mortem study of 14 patients with SARS revealed that 15.9% of patients display histopathological evidence of myocardial infarction. Cases of a stress induced myocarditis such as Takotsubo cardiomyopathy have also been reported during other respiratory viral infections. Cardiac magnetic resonance (CMR) provides an excellent opportunity to non-invasively interrogate the myocardium for a range of pathological abnormalities. The high sensitivity and specificity of CMR for diagnosing cardiac diseases could help establish the underlying cause of myocardial injury in COVID-19 patients. Right ventricular dysfunction is a frequent finding in patients with COVID-19 both acutely and in months after discharge. Abnormalities in the right heart are believed to reflect increased pressures within the pulmonary circulation. Transthoracic echocardiography (TTE) has the ability to definitively detect an increased right heart pressure and is widely used for the assessment of pulmonary hypertension and its causes (e.g., diastolic dysfunction). It is currently unknown whether right heart pressures or function normalise in patients who recover from COVID-19 or whether this remains abnormal months after the acute illness is overcome. The assessment of both myocardial tissue characteristics on CMR and chamber pressures on TTE could accelerate the development of cardiacprotective pathways and treatment options intended to alleviate myocardial injury during respiratory viral epidemics. From a prognostic point of view, the presence and extent of myocardial fibrosis or scar, as detected on CMR, may be helpful in stratifying risks in COVID-19 patients and could identify those in need of more long-term cardiac support (e.g., those with extensive myocardial injury or patients with asymptomatic pre-existing cardiac diseases).

Cardiac Subset (Oxford Only)

We will perform cardiac MRI in patients recruited to cohort B who will be scanned in Oxford, to determine whether HPX-pMRI Chest abnormalities are associated with cardiac impairment, which may potentially contribute to breathlessness.

Expected clinical value of this research

COVID-19 is a public health emergency of global proportions. We expect that the proposed work will provide important and novel insights into persistent exertional dyspnoea in COVID-19 patients. These findings will aid the development of treatment pathways intended to prevent and limit persistent exertional dyspnoea in these patients and potentially in future viral pandemics. A deeper understanding of the impact of COVID-19 on the quality of life of individuals and potential need for ongoing medical surveillance will also be realised through this study.

Outcome Measures Objectives Timepoint(s) Primary To evaluate, using HPX-pMRI a) The detection of a) Baseline scan whether unexplained diffusion and or perfusion exertional dyspnoea in NHLC defects is due to lung damage b) The determination b) Baseline and whether detected follow-up scan abnormalities correlate with symptoms of breathlessness c) The degree of c) End of study pulmonary damage and change detected on follow up scanning Secondary To determine if there are The detection of cardiac Baseline scan **MRI** abnormalities cardiac abnormalities associated with HPX-pMRIdetected abnormalities or present alone as a cause for NHLC patient breathlessness

6. OBJECTIVES AND OUTCOME MEASURES

7. STUDY DESIGN

This is a multi-centre prospective, observational, cohort study comprising of 3 patient cohorts. The study will recruit patients in Cohort A & B through hospital based Long-COVID clinics and Cohort C will be recruited via advertising. Participants attending Long-COVID clinics with a normal CXR will be identified

by the Clinician and informed of the study. If they agree to be contacted, a member of the research team will contact them, explain the study, and assess eligibility. They will then invite them for a baseline visit.

Cohort C will be recruited using posters placed locally (as per our standard practice for local recruitment), A member of the research team will explain the study and invite them for their visit.

The participants will be invited to undertake their baseline study visit at one of the 4 recruiting centres, Oxford, Sheffield, Cardiff or Manchester. At the baseline visit participants provide consent (unless obtained prior to baseline visit), their demographics, past medical history and smoking history will be collected. If necessary, their renal function will be measured according to local hospital policy. They will have their heart rate, blood pressure and oxygen saturations recorded. They will undertake lung function testing (unless testing has been performed within 6 weeks prior to baseline scan), and either a 1 minute sit-to-stand test, or a 6 minute walk test where appropriate. They will also be asked to complete a set of questionnaires.

A low dose Chest CT scan and HPX-pMRI Chest will be performed either at their baseline visit (unless CT scan has been performed within 6 weeks prior to baseline visit) at Oxford and Sheffield. Sheffield will be responsible for implementing a standardised imaging protocol across the two sites and for the central analysis of the MRI data. Patients from Cardiff and Manchester will therefore travel for the MRI scanning at either Oxford or Sheffield following referral pathways that have been tested in previous studies.. If the participant has already had a low dose Chest CT (as per the British Thoracic Society Guidelines for patients with Long COVID and breathlessness), and this was normal or near normal, it will not be repeated.

Following their baseline visit, all participants will be invited to complete up to two additional follow-up visits depending on the outcome of their baseline results, see appendix B).

If an abnormality (a result outside the normal range) is found on the HPX-pMRI Chest scan at baseline the participant will be invited to have all study assessments repeated at visit 2. If at visit 2 further HPX-pMRI abnormalities are found, another follow-up visit, visit 3, will be offered to the participant to have all study assessments repeated again.

If at the baseline visit no abnormalities are found, the participant will be invited to complete the same set of study questionnaires electronically only 3-12 months post their baseline visit.

All of this information will be recorded on the trial database.

8. PARTICIPANT IDENTIFICATION

8.1. Study Participants

Study participants will be identified to join one of the following cohorts, explained in the diagram below:



IRAS:305846

8.2. Inclusion & Exclusion Criteria

Cohort A – Post-hospitalised

Inclusion Criteria

- 1. Aged 18 years or over
- 2. Willing and able to give informed consent
- 3. Previously hospitalised with COVID-19 pneumonia
- 4. Diagnosis of Long-COVID or 'post-COVID syndrome' (as defined by NICE 2020) made through specialised assessment at a designated long-COVID clinic
- 5. Group 1 ONLY:
 - a. More than 3 months post-infection
- 6. Group 2 ONLY:
 - a. 1 10 months post-infection
 - b. Microbiological evidence of COVID-19 infection
 - c. Evidence of architectural distortion on CT

Exclusion Criteria

- 1. Pregnant, lactating or planning pregnancy during the course of the study
- 2. Known chronic renal impairment, with EGFR below 60 mL/min. this will be measured according to local hospital policy.
- 3. Known prior contrast media reaction
- 4. Inability to lie flat for imaging.
- 5. Contraindications to MRI examinations as locally determined.
- 6. Any other reason, as determined by the study investigators, that renders the participant ineligible for the study.
- 7. COVID infection in the last 3 months
- 8. Group 1 ONLY:
 - a. Participant received mechanical ventilation

Cohort B – Non-Hospitalised

Inclusion Criteria

- 1. Aged 18 years or over
- 2. Willing and able to give informed consent
- 3. Group 3 ONLY:
 - a. More than 12 months since infection
 - b. Clinical diagnosis of Long-COVID or 'post-COVID syndrome' (as defined by NICE 2020) made through specialised assessment at a designated Long-COVID clinic
 - c. Significantly breathless upon exertion

4. Group 4 ONLY:

- a. More than 3 months since infection
- b. Microbiological evidence of COVID-19 infection
- c. Clinical diagnosis of Long-COVID or 'post-COVID syndrome' (as defined by NICE 2020) made through specialised assessment at a designated Long-COVID clinic
- d. Significantly breathless upon exertion

5. Group 5 ONLY:

- a. More than 3 months since infection
- b. Microbiological evidence of COVID-19 infection
- c. Clinical diagnosis of Long-COVID or 'post-COVID syndrome' (as defined by NICE 2020) made through specialised assessment at a designated Long-COVID clinic
- d. No respiratory symptoms

Exclusion Criteria

- 1. Admission to hospital with COVID-19 pneumonia
- 2. Pregnant, lactating or planning pregnancy during the course of the study
- Known significant current or prior cardiorespiratory disease as determined at the post-COVID clinic, including significant smoking history (>10 pack years)
- 4. Known chronic renal impairment, with EGFR below 60 mL/min this will be measured according to local hospital policy.
- 5. Known prior contrast media reaction
- 6. Inability to lie flat for imaging.
- 7. Contraindications to MRI examinations as locally determined.
- 8. COVID infection in the last 3 months
- 9. Any other reason, as determined by the study investigators, which renders the participant ineligible for the study.

Cohort C – Controls

Inclusion Criteria

- 1. Aged 18 years or over
- 2. Willing and able to give informed consent
- Microbiological evidence of COVID-19 infection at any time since March 2020 confirmed verbally by the participant
- 4. No on-going symptoms post COVID infection

Exclusion Criteria

1. Admission to hospital with COVID-19 pneumonia

- 2. Pregnant, lactating or planning pregnancy during the course of the study
- 3. Known significant current or prior cardiorespiratory disease including significant smoking history (>10 pack years)
- 4. Known chronic renal impairment, with EGFR below 60 mL/min this will be measured according to local hospital policy.
- Symptoms that fit with the definition of Long-COVID or 'Post-COVID syndrome' as defined by NICE 2020.
- 6. Known prior contrast media reaction
- 7. Inability to lie flat for imaging.
- 8. Contraindications to MRI examinations as locally determined.
- 9. COVID infection in the last 3 months
- 10. Any other reason, as determined by the study investigators, that renders the participant ineligible for the study.

9. PROTOCOL PROCEDURES

9.1. Recruitment

Sites involved in this study will be secondary care service providers in the NHS. Participants for Cohorts A & B will be recruited from Long-COVID clinics and for Cohort C via advertising.

9.2. Screening and Eligibility Assessment

For Cohort A and B, patients will be assessed by the clinical teams for eligibility against the inclusion or exclusion criteria. A member of the research team will contact the participant by telephone to discuss suitability for MRI scan. They may enquire if the potential participant has had a blood test in the last 3 months to check for kidney function – if they have not, they will be asked to have a blood test either at a GP or a hospital drive-through clinic.

The clinical teams at the recruiting sites will confirm that patients have been infected with COVID-19. Pre-screening for eligibility will take place prior to consent. It will be performed by the direct care team using patient notes. There will be no access to patient identifiable data outside of the direct care team prior to consent.

For cohort C, individuals responding to advertisements will be contacted by a member of the Research team, and will be screened to assess their suitability for inclusion by a telephone call which will run through the inclusion and exclusion criteria outlined earlier. They may be asked if they have had a blood test in the last 3 months to check for kidney function – if they have not, they will be asked to have a blood test either at a GP or a hospital drive-through clinic.

Participants must satisfy all the approved inclusion and exclusion criteria of the protocol. Participants will receive the Patient Information Sheet via email, or by post. They will then be invited to participate and informed consent will be obtained at the baseline visit, or at a visit prior to baseline scanning.

9.3. Informed Consent

The participant must personally sign and date the latest approved version of the Informed Consent form before any study specific procedures are performed.

Written and verbal versions of the Participant Information and Informed Consent will be presented to the participants detailing no less than: the exact nature of the study; what it will involve for the participant; the implications and constraints of the protocol; the known side effects and any risks involved in taking part. It will be clearly stated that the participant is free to withdraw from the study at any time for any reason without prejudice to future care, without affecting their legal rights, and with no obligation to give the reason for withdrawal.

The participant will be allowed as much time as wished to consider the information, and the opportunity to question the Investigator, their GP or other independent parties to decide whether they will participate in the study. Written Informed Consent will then be obtained by means of participant-dated signature and dated signature of the person who presented and obtained the Informed Consent. The person who obtained the consent must be suitably qualified and experienced and have been authorised to do so by the Chief/Principal Investigator. A copy of the signed Informed Consent will be given to the participant.

Copies of the consent form will be retained at the study site, and will be stored with the participant hospital record for Cohort A and B.

9.4. Randomisation

This study is not randomised. At registration, all participants will be assigned a unique study ID.

9.5. Blinding and code-breaking

There is no blinding in the study, and no code breaking procedure.

9.6. Description of study intervention(s), comparators and study procedures (clinical)

This is a prospective, observational, cohort study of patients following COVID-19 infection.

9.6.1 Description of study intervention(s)

There will be no study intervention. This is an observational study.

9.6.2 Description of comparator(s)

There is no comparator in this study.

9.6.3 Description of study procedure(s):

Demographics, past medical and smoking (10 mins)

Medical history, demographics, smoking, allergies, medications and anthropometric measurements including height and weight, will be recorded during the study visit.

Observations (10-15 mins)

Blood pressure, heart rate and oxygen saturations

Renal Function Test (5 mins)

If local hospital policy calls for it, some participants will be required to have a blood test to check kidney function within 3 months of their scan.

Lung Function Test (30 Mins)

This will only be performed if not already done so in the 6 weeks prior to visit 1. Pulmonary lung function tests: Spirometry (FEV1 and FVC) and gas transfer (TLCO/KCO).

Exercise Test (10 mins)

A six-minute walk test will be performed where possible, otherwise a one-minute sit-to-stand test combined with pre and post oxygen saturations and mBORG score will be performed.

Questionnaires (45 mins)

Participant to complete questionnaires addressing dyspnoea, fatigue, quality of life and anxiety (Dyspnoea-12, Nijmegen Questionnaire, FACIT-F, GAD-7 and PHQ-9, Visual analogue fatigue rating, work and social adjustment scale and EQ-5D-5L.)

Low dose CT scan (15 mins)

All participants will have a low dose CT scan as part of the trial protocol, if this has not already been performed as part of their routine clinical care within the previous 6 weeks. Patients recruited for cohort A may have had a low dose CT scan as per current recommendations for unexplained post-COVID breathlessness, and this must be normal or nearly normal for them to be recruited (patients with post-COVID-19 ILD are being recruited into UKILD-PCF – Group 2, Cohort A). Cohorts B and C will have low dose CT scans as research scans. Only Group 2 from Cohort A will have a repeat CT scan, at 9 – 15 months post COVID-19 infection. Low dose CT has been shown to reduce radiation exposure, and has been approved previously by the HRA in healthy trial volunteers: the dose will be kept at or below previously HRA-approved exposure.

HPX-pMRI or HPX-MRI (approximately 30 mins)

We will perform hyperpolarised xenon MRI imaging on all enrolled participants, who will be asked to inhale hyperpolarised xenon for one breath hold, on up to 4 occasions per visit (this will enable different data to be acquired, and to allow scanning to be repeated if there is a technical failure). We will include a perfusion MRI (pMRI) using gadolinium contrast to detect areas of alveolar capillary thrombosis. Perfusion MRI has been shown to provide a safe non-ionising means of assessment of capillary perfusion and to be of equal sensitivity to SPECT perfusion imaging. Undertaken alongside dissolved phase xenon imaging, HPX, it will provide a means of assessment of the perfusion deficit in the capillaries and help establish whether breathlessness results from VQ mismatch, interstitial diffusion limitation or a combination of both. This will mean that we will be able to provide a subjective and quantitative method of assessing ventilation, diffusion and perfusion in NHLC patients.

Where a patient is contra-indicated to gadolinium contrast, or has opted out in the consent form, we will perform an HPX-MRI without perfusion. Any participant who is found to have abnormality on HPX-pMRI Chest will be followed up 3-6 months after baseline to repeat all the above assessments. If the HPX-pMRI Chest is normal, the participant will not have follow on imaging or any other tests. MRI scan protocols will be performed as per national agreed protocols for HPX-pMRI Chest in COVID-19 to enable combined data analysis and comparison across different studies.

Hyperpolarised 129Xe

Xenon is an inert gas that is present in the atmosphere in low concentrations. It has no smell or taste. Breathing xenon or undergoing an MRI scan does not involve exposure to ionising radiation. The xenon gas is manufactured and released in MHRA regulatory approved facilities at both Sheffield and Oxford.

Side effects

Xenon has been found to be extremely safe and well tolerated. Common side effects include: dizziness, drowsiness, deepening of the voice and a feeling of euphoria. They typically last less than a minute. In studies in over 1000 patients performed at both Oxford and Sheffield no SAEs have been reported related to xenon inhalation.

Gadolinium contrast

Contrast dyes (Gadolinium) are commonly used in MRI scans for medical care. They may occasionally cause mild rash, headache, and very rarely (less than 1 in 5000) a more severe allergic reaction. Although the risk of a severe reaction is small, there will be trained medical personnel present at all times with access to emergency equipment should this be required. In people with healthy kidneys, this dye is removed from the bloodstream within a few hours. Kidney function will be tested according to hospital policy.

Cardiac MRI (15 mins)

We will also aim to perform cardiac MRI in patients recruited to cohort B for scanning in Oxford, to determine whether HPX-pMRI Chest abnormalities are associated with cardiac impairment, which may potentially contribute to breathlessness. Cardiac MRI sequences will be brief and restricted to assessment of biventricular function, strain and strain rate.

9.7. Baseline and follow up assessments

Initial patient approach

Eligibility assessment by clinical team Study information provided to eligible patients

Visit 1 Baseline (Participants post COVID-19 infection)

- 1. Review of eligibility with participant
- 2. Obtain written informed consent (unless obtained prior to baseline visit)

All procedures will be performed as listed above and as detailed in appendix B .

<u>Visit 2 and 3</u> (3-12 months post baseline visit) - abnormal HPX-pMRI results at baseline or Visit 2

Participants will be invited to attend for follow up where all assessments from visit 1 will be repeated.

Visit 2 and 3 (3-12 months post baseline visit) - normal results at baseline or at Visit 2

Participants will be invited for remote follow up and asked to complete the same set of questionnaires as at visit 1.

Face to face visits will be scheduled to take place on 1 day – they will take approximately half the day.

9.8. Early Discontinuation/Withdrawal of Participants

During the course of the study a participant may choose to withdraw early from the study treatment at any time. This may happen for several reasons, including but not limited to:

- The occurrence of what the participant perceives as an intolerable AE
- Inability to comply with study procedures
- Participant decision

According to the design of the study, participants have the following options if they withdraw from the study:

- 1) Participants can withdraw from the study but permit data obtained up until the point of withdrawal to be retained for use in the study analysis. No further data would be collected after withdrawal.
- 2) Participants can withdraw from study scans but continue with data collection.

In addition, the Investigator may discontinue a participant from the study at any time if the Investigator considers it necessary for any reason including, but not limited to:

- Pregnancy
- Ineligibility (either arising during the study or retrospectively having been overlooked at screening)
- Clinical decision

Participants that choose to withdraw from the study will not be replaced. The type of withdrawal and reason for withdrawal will be recorded in the CRF. If the participant is withdrawn due to an adverse event, the Investigator will arrange for follow-up visits or telephone calls until the adverse event has resolved or stabilised.

9.9. Definition of End of Study

The end of study will be defined as when all data has been entered and all queries resolved.

10. SAFETY REPORTING

10.1. Definition of Serious Adverse Events

A serious adverse event is any untoward medical occurrence that:

- results in death
- is life-threatening
- requires inpatient hospitalisation or prolongation of existing hospitalisation
- results in persistent or significant disability/incapacity
- consists of congenital anomaly or birth defect

Other 'important medical events' may also be considered a serious adverse event when, based upon appropriate medical judgement, the event may jeopardise the participant and may require medical or surgical intervention to prevent one of the outcomes listed above. NOTE: The term "life-threatening" in the definition of "serious" refers to an event in which the participant was at risk of death at the time of the event; it does not refer to an event which hypothetically might have caused death if it were more severe.

10.2. Reporting Procedures for Serious Adverse Events

A serious adverse event (SAE) occurring to a participant should be reported to the REC that gave a favourable opinion of the study where in the opinion of the Chief Investigator the event was 'related' (resulted from administration of any of the research procedures) and 'unexpected' in relation to those procedures. Reports of related and unexpected SAEs should be submitted within 15 working days of the Chief Investigator becoming aware of the event, using the HRA report of serious adverse event form (see HRA website). However, CT and Xenon MRI are widely used and we have established data on safety, therefore SAEs or related AEs are not expected.

11. STATISTICS AND ANALYSIS

11.1. Description of the Statistical Methods

All baseline characteristics and measures will be summarised overall and by group, using appropriate summary statistics.

We will describe the prevalence, quantitative and qualitative extent of persistent lung and cardiacdamage, in patients with COVID-19 disease and compare this with our matched controls. We will examine the relationship between extent of lung and cardiac injury, functional impairment and adverse clinical outcomes. We will compare quality of life and mental health scores in post SARS-CoV-2 infected patients and matched controls and assess its relationship with extent of lung and cardiac damage and initial hospital admission.

Number and proportions with HPX-pMRI Chest abnormalities will be summarised by group in order to estimate the sensitivity and specificity of the procedure. We will estimate the 95% confidence intervals for sensitivity and specificity. We will use logistic regression to build a predictive model of the relationship between presence of abnormalities and group with the aim of determining whether HPX-pMRI Chest abnormalities can discriminate between patients and controls. Mean Xenon detected gas transfer (RBC:TP) will be summarised by group and analyse using a linear regression model to estimate the mean difference between groups.

Numbers and proportions of participants with cardiac abnormalities detected will be summarised by group and presence of HPX-pMRI Chest abnormalities. Binary and categorical outcomes will be analysed using logistic regression.

Quality of life, mental health scores and frailty measures will be summarised by group and time point using mean and standard deviation (or median and interquartile range if non normally distributed).

Continuous measures, such as questionnaires, will be analysed using linear mixed models with fixed effects for group and timepoint (baseline visit and any follow up visits), as well as a group x timepoint interaction in order to determine the timepoint specific effects. A participant specific random effect will be fitted to account for clustering of timepoints within participants.

All models will be adjusted for baseline characteristics thought to be prognostic.

11.2. Sample Size Determination

Our previous data in patients enrolled in C-MORE POST⁻ a sub-study of the HRA approved C-MORE study - demonstrated significant abnormalities in 18/22 scanned hospitalised patients. While we would not expect as high a proportion of abnormality within an NHLC patient cohort, the POST scan results provide important data on which to plan the sample size.

A sample size of 100 for the Cohort A is proposed, as this should be sufficient to perform multivariate analysis on the limited feature sets, and will be sufficient to assess the utility of hyperpolarised xenon to detect alterations if present in gas exchange in post-hospitalised patients.

Our planned sample size in Cohort B is 200 patients with NHLC and symptoms of breathless, 50 NHLC patients with no symptoms of breathlessness, and in Cohort C, 50 controls. Assuming a lower sensitivity of 70% in practice, then calculating for a 95% confidence interval and a width of 0.131 (equivalent to a lower limit of 63.1%) would require a sample size of 200 patients. If specificity is assumed to be 90%, then a sample size of 50 controls would give a confidence interval of width 0.185 (corresponding to a lower limit of 78.2%).

The mean value of Xenon detected gas transfer (RBC:TP) in patients was 0.3 (0.1) versus 0.6 (0.3) in healthy controls (absolute difference 0.3). This provides 90% power to detect a difference of 0.094 (based on a mean of 0.3 and SD of 0.1 in the control group and a SD of 0.3 in the Long-Covid group) in this key MRI parameter, which is well below the difference detected in POST. The level of statistical significance is set at 1.7% to allow for multiple comparisons between both patients and controls and patients with and without breathlessness.

11.3. Analysis populations

All participants who have completed HPX-pMRI scans will be included in the analyses.

11.4. Dissemination of Findings

Given the priority for dissemination of findings to inform public health and government bodies, interim descriptive results will be posted online after the first 50 participants then every 100 participants thereafter at the earliest time point of assessment.

11.5. Stopping rules

Not applicable.

11.6. The Level of Statistical Significance

All tests are two-tailed and p-values <0.05 were considered significant.

11.7. Procedure for Accounting for Missing, Unused, and Spurious Data.

The analysis models are valid under a missing at random (MAR) assumption. Baseline demographic variables will be summarised by availability of outcomes and any variables that appear to be predictive of missingness will be included in the analysis models in a sensitivity analysis. We do not expect any data to be missing at baseline but in the case of substantial (>10%) missing data at baseline multiple imputation including any variables predictive of missingness will be carried out as a sensitivity analysis.

For continuous measures outliers will be defined as any value that lies more than two standard deviations from the mean. We will explore the effect of excluding any outliers from the analysis in a sensitivity analysis. If the estimates do not differ substantively then the model retaining the outliers will be reported.

11.8. Procedures for Reporting any Deviation(s) from the Original Statistical Plan

Any deviations from the statistical analysis plan will be detailed along with the reason for the deviation in the statistical analysis report.

11.9. Health Economics Analysis

Not applicable.

12. DATA MANAGEMENT

The CT scans will be entered onto the hospital image archiving system, PACS, so they can be clinically reviewed and available for patient care by clinical teams. All image data will be pseudonymised at point of acquisition and transferred to Oxford University Hospitals NHS FT where it will be held on a secure server in the Oxford Radiology Research Unit, designed to hold information securely and in accordance with legal requirements. A copy of all pseudonymised MR images from sites will be shared with the

University of Sheffield, and will be held on a secure XNAT server at the University of Sheffield, which is the main site for analysis, and will contain no personal data.

All data transfer and data use will be in concordance with data protection regulation, Trust information governance procedures, in addition to the ethical and governance standards in the University of Oxford and the University of Sheffield. Access to the pseudonymised data will only be available to the relevant study teams and clinicians involved in the project.

The de-identified data from Oxford University Hospitals NHS FT may also be made available to third parties, as companies are working with the investigators to improve HPX-pMRI Chest imaging for patients with Long COVID and other lung diseases, subject to appropriate data sharing agreements.

All other pseudonymised data will be captured electronically using an electronic case report form (eCRF), Castor EDC, that meets data security and regulatory requirements (compliant to ISO27001). According to Castor EDC, all data is stored on servers hosted by certified hosting parties. In the UK, the servers are hosted by Microsoft Azure. Backups are made twice per day and moved to a separate geographic location daily. ORRU holds access to the study profile and can grant access for data entry only to the appropriate sites.

12.1. Source Data

Source documents are where data are first recorded, and from which participants' CRF data are obtained. These include, but are not limited to, hospital records (from which medical history and previous and concurrent medication may be summarised into the CRF), clinical and office charts, laboratory and pharmacy records, diaries, microfiches, radiographs, and correspondence.

CRF entries will be considered source data if the CRF is the site of the original recording (e.g. there is no other written or electronic record of data). All documents will be stored safely and securely in confidential conditions. On all study-specific documents, other than the signed consent, the participant will be referred to by the study participant number/code, not by name, with the exception of the CRF, where participant initials may be added.

12.2. Access to Data

Direct access will be granted to authorised representatives from the Sponsor and host institution for monitoring and/or audit of the study to ensure compliance with regulations.

12.3. Data Recording and Record Keeping

Study data will be recorded in a pseudonymised manner and processed electronically where applicable; participants would not be identifiable from this. All participants will be identified by a code number (Study ID number) on electronic case report forms and any electronic databases i.e. their identity will remain unknown. Electronic cloud based software such as Castor EDC will be used to collect the study data.

This is an electronic data capture and management system that permits secure multi-site access. The server is based in the UK and complies with relevant laws to ensure that the data is held securely.

All source documents and questionnaires will have all identifiers removed and replaced by only the study ID number.

Those documents retaining personal identifiable data (e.g informed consent forms) will be stored securely in lockable cabinets and are only accessible by study staff and authorised personnel.

Personal data such as contact details and information that could identify a participant, (except for those who have consented to be approached for future research) will be destroyed as soon as it is practical to do so and no later than 12 months after the end of the study. During the study, the hard copy study data will be stored securely and at the end of the study, this will be archived in a secure commercial archive location with restricted access.

12.4 Data Sharing

Where the clinical team has verbal agreement from patients to share their contact details with the research team these contact details will be shared securely for example through NHS email accounts.

Pseudonymised imaging data will be shared from research sites to the sponsor via the Oxford University Hospitals NHS Foundation Trust secure XNAT server. This data will also be shared with Sheffield University for the purposes of analysis. There is a data transfer agreement in place detailing the relevant flows.

13. QUALITY ASSURANCE PROCEDURES

The study may be monitored, or audited in accordance with the current approved protocol, GCP, relevant regulations and standard operating procedures.

Data collected for the study may be reviewed for auditing and monitoring by authorised persons from the sponsor, regulatory authorities or their local host institutions to make sure that study is being carried out correctly. All investigators have a duty of confidentiality to research participants and nothing that could reveal their identity would be disclosed outside the research study team or site without specific consent from the research participant.

13.1. Risk assessment

A risk assessment and monitoring is not deemed necessary. Any risks associated with the study assessments or data management have been outlined in the protocol (sections 13 and 17.3) and mitigations outlined.

13.2. Study Monitoring

No monitoring required for this study.

13.3. Study Committees

13.3.1 Trial Management Group

Trial Management Group will meet regularly throughout the trial to discuss the day-to-day management of the trial. A TMG charter will be written detailing all the requirements.

13.3.2 Trial Steering Committee

A TSC will be convened to keep oversight of the trial. A charter will be written explaining the role of the TSC and each of its members. All members are required to sign a declaration of their participation. The charter will define how often the committee will meet during the course of the study.

13.3.3 ORTU Safety Oversight Group (SOG)

The Oxford Respiratory Trials Unit (ORTU) will conduct a review of all SAEs reported during the reporting period and cumulatively. The aims of this committee include:

- To pick up any trends, such as increases in un/expected events, and take appropriate action
- To seek additional advice or information from investigators where required
- To evaluate the risk of the trial continuing and take appropriate action where necessary

14. PROTOCOL DEVIATIONS

A study related deviation is a departure from the ethically approved study protocol or other study document or process (e.g. consent process or administration of study intervention) or from Good Clinical Practice (GCP) or any applicable regulatory requirements. Any deviations from the protocol will be documented in a protocol deviation form and filed in the study master file.

15. SERIOUS BREACHES

A "serious breach" is a breach of the protocol or of the conditions or principles of Good Clinical Practice which is likely to affect to a significant degree –

- (a) the safety or physical or mental integrity of the trial subjects; or
- (b) the scientific value of the research.

In the event that a serious breach is suspected the Sponsor must be contacted within 1 working day. In collaboration with the C.I., the serious breach will be reviewed by the Sponsor and, if appropriate, the Sponsor will report it to the approving REC committee and the relevant NHS host organisation within seven calendar days.

16. ETHICAL AND REGULATORY CONSIDERATIONS

16.1. Declaration of Helsinki

The Investigator will ensure that this study is conducted in accordance with the principles of the Declaration of Helsinki. Guidelines for Good Clinical Practice. The Investigator will ensure that this study is conducted in accordance with relevant regulations and with Good Clinical Practice.

16.2. Approvals

Following Sponsor approval, the protocol, the informed consent form, participant information sheet and any proposed advertising material will be submitted to an appropriate Research Ethics Committee (REC), and HRA (where required) and host institutions for written approval.

The Investigator will submit and, where necessary, obtain approval from the above parties for all substantial amendments to the original approved documents.

16.3. Other Ethical Considerations

If any significant incidental unexpected abnormality is identified from blood tests or scanning, the participants primary or secondary care doctor will be informed dependent upon their route of enrolment.

16.4. Reporting

The CI shall submit once a year throughout the study, or on request, an Annual Progress report to the REC Committee, HRA (where required), host organisation, Sponsor and funder (where required). In addition, an End of Study notification and final report will be submitted to the same parties.

16.5. Transparency in Research

Prior to the recruitment of the first participant, the trial will have been registered on a publicly accessible database such as ISRCTN.

16.6. Participant Confidentiality

The study will comply with the UK General Data Protection Regulation (UK GDPR) and Data Protection Act 2018, which require data to be anonymised as soon as it is practical to do so. The processing of the personal data of participants will be minimised by making use of a unique participant study number only on all study documents and any electronic database(s). All documents will be stored securely and only accessible by study staff and authorised personnel. The study staff will safeguard the privacy of participants' personal data.

16.7. Expenses and Benefits

Reasonable travel expenses for any visits additional to normal care will be reimbursed on production of receipts, or a mileage allowance provided as appropriate. There will be no direct benefit to participants,

although there maybe potential benefit if previously unknown health related findings were identified as part of the research assessments.

17. FINANCE AND INSURANCE

17.1. Funding

This study is funded by NIHR, the University of Oxford, the National Consortium of Intelligent Medical Imaging.

17.2. Insurance

The University has a specialist insurance policy in place which would operate in the event of any participant suffering harm as a result of their involvement in the research (Newline Underwriting Management Ltd, at Lloyd's of London).

17.3. Contractual arrangements

Appropriate contractual arrangements will be put in place with all third parties. Site agreements will be put in place with NHS sites.

18. PUBLICATION POLICY

The Investigators will be involved in reviewing drafts of the manuscripts, abstracts, press releases and any other publications arising from the study. Authors will acknowledge that the study was funded by the NIHR, NCIMI and other funding bodies that will continue to fund this study. Authorship will be determined in accordance with the ICMJE guidelines and other contributors will be acknowledged.

19. DEVELOPMENT OF A NEW PRODUCT/ PROCESS OR THE GENERATION OF INTELLECTUAL PROPERTY

Ownership of IP generated by employees of the University vests in the University. The University will ensure appropriate arrangements are in place as regards any new IP arising from the trial.

20. ARCHIVING

At the end of the study research files will be archived for 5 years and in line with local SOPs.

21. REFERENCES

1. Kannan S, ALI PSS, Sheeza A, Hemalatha K. COVID-19 (Novel Coronavirus 2019)–recent trends. European Review for Medical and Pharmacological Sciences. 2020;24:2006-11.

2. Dong E, Du H, Gardner L. An interactive web-based dashboard to track COVID-19 in real time. The Lancet Infectious Diseases. 2020.

3. Richman DD, Whitley RJ, Hayden FG. Clinical virology: John Wiley & Sons; 2016.

4. Ksiazek TG, Erdman D, Goldsmith CS, Zaki SR, Peret T, Emery S, et al. A novel coronavirus associated with severe acute respiratory syndrome. New England journal of medicine. 2003;348(20):1953-66.

5. Zhou F, Yu T, Du R, Fan G, Liu Y, Liu Z, et al. Clinical course and risk factors for mortality of adult inpatients with COVID-19 in Wuhan, China: a retrospective cohort study. The Lancet. 2020.

6. Wu C, Chen X, Cai Y, Zhou X, Xu S, Huang H, et al. Risk Factors Associated With Acute Respiratory Distress Syndrome and Death in Patients With Coronavirus Disease 2019 Pneumonia in Wuhan, China. JAMA Internal Medicine.

7. Shi H, Han X, Jiang N, Cao Y, Alwalid O, Gu J, et al. Radiological findings from 81 patients with COVID-19 pneumonia in Wuhan, China: a descriptive study. The Lancet Infectious Diseases. 2020.

8. Guan W-j, Ni Z-y, Hu Y, Liang W-h, Ou C-q, He J-x, et al. Clinical characteristics of coronavirus disease 2019 in China. New England Journal of Medicine. 2020.

9. Pan F, Ye T, Sun P, Gui S, Liang B, Li L, et al. Time course of lung changes on chest CT during recovery from 2019 novel coronavirus (COVID-19) pneumonia. Radiology. 2020:200370.

10. Calverley PM. The GOLD classification has advanced understanding of COPD. American journal of respiratory and critical care medicine. 2004;170(3):211-2.

11. European RS, Society AT. American Thoracic Society/European Respiratory Society international multidisciplinary consensus classification of the idiopathic interstitial pneumonias. This joint statement of the American Thoracic Society (ATS), and the European Respiratory Society (ERS) was adopted by the ATS board of directors, June 2001 and by the ERS executive committee, June 2001. American Journal of Respiratory and Critical Care Medicine. 2002;165(2):277.

12. Huang C, Wang Y, Li X, Ren L, Zhao J, Hu Y, et al. Clinical features of patients infected with 2019 novel coronavirus in Wuhan, China. The Lancet. 2020;395(10223):497-506.

13. Wang D, Hu B, Hu C, Zhu F, Liu X, Zhang J, et al. Clinical characteristics of 138 hospitalized patients with 2019 novel coronavirus–infected pneumonia in Wuhan, China. Jama. 2020.

14. Mao L, Wang M, Chen S, He Q, Chang J, Hong C, et al. Neurological Manifestations of Hospitalized Patients with COVID-19 in Wuhan, China: a retrospective case series study. medRxiv. 2020:2020.02.22.20026500.

15. Jiang Y, Xu J, Zhou C, Wu Z, Zhong S, Liu J, et al. Characterization of cytokine/chemokine profiles of severe acute respiratory syndrome. American journal of respiratory and critical care medicine. 2005;171(8):850-7.

16. Edwards S, Small JD, Geratz JD, Alexander LK, Baric RS. An experimental model for myocarditis and congestive heart failure after rabbit coronavirus infection. Journal of Infectious Diseases. 1992;165(1):134-40.

17. Tsui K-L, Leung T-C, Yam LY-C, So LK-Y, Poon E, Lung K-C, et al. Coronary plague instability in severe acute respiratory syndrome. International journal of cardiology. 2005;99(3):471-2.

18. Chong PY, Chui P, Ling AE, Franks TJ, Tai DY, Leo YS, et al. Analysis of deaths during the severe acute respiratory syndrome (SARS) epidemic in Singapore: challenges in determining a SARS diagnosis. Archives of pathology & laboratory medicine. 2004;128(2):195-204.

19. Buzon J, Roignot O, Lemoine S, Perez P, Kimmoun A, Levy B, et al. Takotsubo cardiomyopathy triggered by influenza A virus. Internal Medicine. 2015;54(16):2017-9.

20. Pathik B, Raman B, Mohd Amin NH, Mahadavan D, Rajendran S, McGavigan AD, et al. Troponinpositive chest pain with unobstructed coronary arteries: incremental diagnostic value of cardiovascular magnetic resonance imaging. European Heart Journal-Cardiovascular Imaging. 2016;17(10):1146-52.

21. Szekely Y, Lichter Y, Taieb P, Banai A, Hochstadt A, Merdler I, et al. The Spectrum of Cardiac Manifestations in Coronavirus Disease 2019 (COVID-19)-a Systematic Echocardiographic Study. Circulation. 2020.

22. Zhang C, Shi L, Wang F-S. Liver injury in COVID-19: management and challenges. The Lancet Gastroenterology & Hepatology. 2020.

23. Yeo C, Kaushal S, Yeo D. Enteric involvement of coronaviruses: is faecal–oral transmission of SARS-CoV-2 possible? The Lancet Gastroenterology & Hepatology. 2020.

24. Chau TN, Lee KC, Yao H, Tsang TY, Chow TC, Yeung YC, et al. SARS-associated viral hepatitis caused by a novel coronavirus: report of three cases. Hepatology. 2004;39(2):302-10.

25. Núñez M. Hepatotoxicity of antiretrovirals: incidence, mechanisms and management. Journal of hepatology. 2006;44:S132-S9.

26. Banerjee R, Pavlides M, Tunnicliffe EM, Piechnik SK, Sarania N, Philips R, et al. Multiparametric magnetic resonance for the non-invasive diagnosis of liver disease. Journal of hepatology. 2014;60(1):69-77.

27. Pavlides M, Banerjee R, Sellwood J, Kelly CJ, Robson MD, Booth JC, et al. Multiparametric magnetic resonance imaging predicts clinical outcomes in patients with chronic liver disease. Journal of hepatology. 2016;64(2):308-15.

28. McDonald N, Eddowes PJ, Hodson J, Semple SI, Davies NP, Kelly CJ, et al. Multiparametric magnetic resonance imaging for quantitation of liver disease: a two-centre cross-sectional observational study. Scientific reports. 2018;8(1):1-10.

29. Cheng Y, Luo R, Wang K, Zhang M, Wang Z, Dong L, et al. Kidney disease is associated with inhospital death of patients with COVID-19. Kidney international. 2020.

30. Richardson S, Hirsch JS, Narasimhan M, Crawford JM, McGinn T, Davidson KW, et al. Presenting characteristics, comorbidities, and outcomes among 5700 patients hospitalized with COVID-19 in the New York City area. Jama. 2020.

31. Varga Z, Flammer AJ, Steiger P, Haberecker M, Andermatt R, Zinkernagel AS, et al. Endothelial cell infection and endotheliitis in COVID-19. The Lancet. 2020;395(10234):1417-8.

32. Michael BaE, A. CORONAVIRUS AND ENCEPHALITIS. 2020.

33. Xu J, Zhong S, Liu J, Li L, Li Y, Wu X, et al. Detection of severe acute respiratory syndrome coronavirus in the brain: potential role of the chemokine mig in pathogenesis. Clinical infectious diseases. 2005;41(8):1089-96.

34. Netland J, Meyerholz DK, Moore S, Cassell M, Perlman S. Severe acute respiratory syndrome coronavirus infection causes neuronal death in the absence of encephalitis in mice transgenic for human ACE2. J Virol. 2008;82(15):7264-75.

35. Hui D, Joynt G, Wong K, Gomersall C, Li T, Antonio G, et al. Impact of severe acute respiratory syndrome (SARS) on pulmonary function, functional capacity and quality of life in a cohort of survivors. Thorax. 2005;60(5):401-9.

36. Tisoncik JR, Korth MJ, Simmons CP, Farrar J, Martin TR, Katze MG. Into the eye of the cytokine storm. Microbiology and Molecular Biology Reviews. 2012;76(1):16-32.

37. Slutsky AS, Ranieri VM. Ventilator-induced lung injury. New England Journal of Medicine. 2013;369(22):2126-36.

38. McEvoy CE, Niewoehner DE. Adverse effects of corticosteroid therapy for COPD: a critical review. Chest. 1997;111(3):732-43.

39. Tremblay LN, Slutsky AS. Ventilator-induced injury: from barotrauma to biotrauma. Proceedings of the Association of American Physicians. 1998;110(6):482-8.

40. Raman B, Cassar MP, Tunnicliffe EM, Filippini N, Griffanti L, Alfaro-Almagro F, et al. Medium-term effects of SARS-CoV-2 infection on multiple vital organs, exercise capacity, cognition, quality of life and mental health, post-hospital discharge. EClinicalMedicine. 2021;31:100683.

41. Wong R, Lopaschuk G, Zhu G, Walker D, Catellier D, Burton D, et al. Skeletal muscle metabolism in the chronic fatigue syndrome: in vivo assessment by 31P nuclear magnetic resonance spectroscopy. Chest. 1992;102(6):1716-22.

42. Valkovič L, Chmelík M, Krššák M. In-vivo 31P-MRS of skeletal muscle and liver: a way for non-invasive assessment of their metabolism. Analytical biochemistry. 2017;529:193-215.

43. Stoll VM, Hess AT, Rodgers CT, Bissell MM, Dyverfeldt P, Ebbers T, et al. Left ventricular flow analysis: novel imaging biomarkers and predictors of exercise capacity in heart failure. Circulation: Cardiovascular Imaging. 2019;12(5):e008130.

44. Huckstep OJ, Williamson W, Telles F, Burchert H, Bertagnolli M, Herdman C, et al. Physiological stress elicits impaired left ventricular function in preterm-born adults. Journal of the American College of Cardiology. 2018;71(12):1347-56.

45. Grist JT, Chen M, Collier GJ, Raman B, AbuEid G, McIntyre A, Matthews V, Fraser E, Ho LP, Wild JM, Gleeson F. Hyperpolarized 129Xe MRI Abnormalities in Dyspneic Participants 3 Months after COVID-19 Pneumonia: Preliminary Results. Radiology. 2021 May 25:210033. doi: 10.1148/radiol.2021210033. **46.** Min L et al.

22. APPENDIX A: SUPPORTING DATA



POST Study Follow up HPX-MRI data

- HPX-MRI RBC/TP data from the POST study Showing:
- 5 controls with normal RBC/TP lung diffusion
 - 9 patients from POST

 At baseline 3 months post discharge
 At 6 month follow up

 This shows that the majority 6/9 of the
 - patients are not improving
 - The outlier has clearly improved,
 - and was the least abnormal on baseline scanning



Figure 2: HPX-MRI scans in 2 NHLC patients.

The top row is a patient with ongoing breathlessness and the bottom row in a patient with improved breathlessness

RBC:TP

RBC:TP

Cohort A	Hospitalised	Hospitalised	Hospitalised	Hospitalised
		Participants	Participants	Participants
	All participants	invited only if	invited only if	invited only if
	invited	abnormal results	normal results at	abnormal results
		at baseline	baseline	at V2
Vicit	Visit 1	Visit 2 (Follow-	Visit 2 (Follow-	Visit 3 (Follow-
VISIL	(Baseline)	Up)	Up) remote	Up)
Period	Day 0	6 months	3-12 months	12 months
		(-/+ 3 months)		(-/+ 3 months)
Eligibility assessment	~			
Informed consent	✓ †			
Demographics, medical history, concomitant				
medications				
Cardiorespiratory examination (incl. observations,				
heart rate, blood pressure and oxygen saturations)				· ·
Blood test (Renal function)	✓ ^{††}	✓ ^{††}		✓ ^{††}
Questionnaires	~	~	~	~
Lung Function Tests	✓*	✓*		✓*
Excercise test	~	~		~
Low Dose CT	✓*	✔**		
MRI Chest	~	~		~
MRI Chest with 129-Xenon gas	~	~		~
MRI Chest with gadolinium contrast injection	~	~		~

23. APPENDIX B: SCHEDULE OF STUDY PROCEDURES

[†]Where appropriate, informed consent may take place prior to the baseline visit

¹⁺ In some locations, participants will have renal function testing performed according to local hospital policy. *LFT and low dose CT to be performed only if they have not been performed clinically within 6 weeks prior to scan**Follow-up CT scan performed on Group 2 only.

Cohort B	Non-hospitalised	Non-hospitalised	Non-hospitalised	Non-hospitalised
		Participants	Participants	Participants
	All participants	invited only if	invited only if	invited only if
	invited	abnormal results	normal results at	abnormal results
		at baseline	baseline	at V2
Vicit	Visit 1	Visit 2 (Follow-	Visit 2 (Follow-	Visit 2 (Follow-
VISIL	(Baseline)	Up)	Up) remote	Up) remote
Period	Day 0	6 months	3-12 months	12 months
T enou	Day 0	(-/+ 3 months)	5- 12 11011013	(-/+ 3 months)
Eligibility assessment	~			
Informed consent	✓ †			
Demographics, medical history, concomitant	~			
medications				
Cardiorespiratory examination (incl. observations,	~	4		4
heart rate, blood pressure and oxygen saturations)				
Blood test (Renal function)	✓**	✓**		✓ **

Questionnaires	v	~	v	v
Lung Function Tests	✔*	✓*		✓*
Exercise test	V	~		~
Low Dose CT	✔*			
MRI Chest	V	~		~
MRI Cardiac ^{††}	V	~		~
MRI Chest with 129-Xenon gas	V	~		~
MRI Chest with gadolinium contrast injection	V	~		~

[†]Where appropriate, informed consent may take place prior to the baseline visit

⁺⁺ Cardiac MRI to be performed in Oxford only

*LFT and low dose CT to be performed only if they have not been performed clinically within 6 weeks prior to scan** In some locations, participants will have renal function testing performed according to local hospital policy.

Cohort C	Controls
Visit	Visit 1
Period	Day 0
Eligibility assessment	~
Informed consent	✓*
Demographics, medical history, concomitant medications	v
Cardiorespiratory examination (incl. observations, heart rate, blood pressure and oxygen saturations)	~
Blood test (Renal Function)	✓ [↑]
Lung Function Tests	~
Excercise test	~
Low Dose CT	~
MRI Chest	~
MRI Cardiac	✓**
MRI Chest with 129-Xenon gas	~
MRI Chest with gadolinium contrast injection	~

* Where appropriate, informed consent may take place prior to the baseline visit

** Cardiac MRI to be performed in Oxford only

† In some locations, participants will have renal function testing performed according to local hospital policy.

24. APPENDIX C: AMENDMENT HISTORY

Amendment No.	Protocol Version No.	Date issued	Author(s) of changes	Details of Changes made
1	2.0	30Mar2022	Emma Hedley	7. Removal of advertising email from the protocol.
				12.4 Additional information added to the Data Sharing Section to include that imaging data will be shared with Sheffield University for analysis.
2	3.0	07Apr2022	Emma Hedley	 8.2 clarification of inclusion criteria (Significantly breathless upon exertion) for groups 3, 4 and 5 (No respiratory symptoms) and Cohort C – controls (No on- going symptoms post COVID infection) in-keeping with the flow chart on page 19. Addition of an exclusion criteria "COVID infection in the last 3 months" for all groups. 9.8 Participants can withdraw from study scans but continue with data collection. 10. Removal of the word "an unexpected" in the SAE definition results in death